Author's response to reviews

Title: Trans-visceral migration of retained surgical gauze as cause of intestinal obstruction: a case report.

Authors:

Nello Grassi (NG40147@policlinico.pa.it)
Calogero Cipolla (calogercipolla@virgilio.it)
Adriana Torcivia (adrianaotorcivia@libero.it)
Alessandro Bottino (AB40613@policlinico.pa.it)
Eugenio Fiorentino (efiorentino@unipa.it)
Leonardo Ficano (leoficano@tin.it)
Gianni Pantuso (gianpant@katamail.com)

Version: 2 Date: 20 January 2007

Author's response to reviews: see over
Dear Editors,

We have revised our manuscript: "Trans-visceral migration of Retained Surgical Gauze as Cause of Intestinal Obstruction" (Ms 1049071754128831) according to your suggestions and would like to resubmit it for publication in the "Journal of Medical Case Reports"

We have changed the title as requested from "Trans-visceral migration of Retained Surgical Gauze as Cause of Intestinal Obstruction" to "Trans-visceral migration of Retained Surgical Gauze as Cause of Intestinal Obstruction: a case report"; moreover we reduced the reference list from sixteen to ten references and the figures from four to three.

Thanks for your interest to our manuscript, yours sincerely,
Calogero Cipolla

Corresponding author:
Dr. Calogero Cipolla MD
Dipartimento di Oncologia - Divisione di Chirurgia Generale ed Oncologica
A.O.U. Policlinico “Paolo Giaccone”
Via del Vespro 129, 90127 Palermo (Italy)

Tel +39 091 6554520
Fax +39 091 6554429
calogerocipolla@virgilio.it